Author’s response to reviews

Title: Pediatric First Aid Knowledge and Attitudes among Staff in the Preschools of Shanghai, China.

Authors:

Feng Li (lifeng0228@163.com)
Fan Jiang (fanjiang@shsmu.edu.cn)
Ming xing Jin (zhujing7@msn.cn)
Lan yu Qiu (qiuguiyulan@hotmail.com)
Ming xiao Shen (xmshen@online.sh.cn)

Version: 2 Date: 2 March 2012

Author’s response to reviews:

Dear Editor,

Thank you for your useful comments and suggestions on the language and structure of our manuscript. We have modified the manuscript accordingly, and detailed corrections are listed below point by point:

1. Ethics Statement: Experimental research that is reported in the manuscript must have been performed with the approval of an appropriate ethics committee. Research carried out on humans must be in compliance with the Helsinki Declaration. A statement to this effect must appear in the Methods section of the manuscript, including the name of the body which gave approval.

Answer: We have supplemented this in the Methods section of the manuscript.

2. Copyediting request: After reading through your manuscript, we feel that the quality of written English needs to be improved before the manuscript can be considered further.

Answer: We have modified this paper repeatedly to reduce the grammar mistakes and inaccurate description as far as possible. In addition, the revised and reformulated manuscript has been polished by a fluent English speaking colleague. We believe that the language is now acceptable for the review process.

We look forward to hearing from you and thank for your consideration.

Sincerely yours,

Feng Li, Fan Jiang, Xingming Jin, Yulan Qiu, Xiaoming Shen